Author’s response to reviews

Title: Time-Lapse Angiography of the Ocular Fundus: A New Video-Angiography

Authors:

Chufeng Gu (wts_gchuf@163.com)
Jili Chen (corneachen@163.com)
Tong Su (sutong566@163.com)
Qinghua Qiu (qinghuaqiu@163.com)

Version: 2 Date: 03 Nov 2019

Author’s response to reviews:

Dear Editor,

We are grateful for your appreciation and precise revisions to our manuscript. These comments are helpful for improving my paper. We have studied comments carefully and did further verification to revise our paper. The answers to the reviewer’s comments are as follows:

yang chen (Reviewer 1): This paper looks like a report, lacking statistics analysis, quantitative results. It is hard to reach scientific conclusion without quantitatively statistical analysis.

Response: Thank you for your reminder. We introduced a modified method to dynamically present imaging results of the ocular fundus. Quantitative research compared to other techniques is ongoing, but it will take some time to complete, so we wanted to share this technology with you first. In this technical advance article, we mainly introduced the principle, method, and result of this technique.

Also the language still needs to be improved, some example: "eyes moving and blinking", "time moves"

Response: Thank you for your reminder. American Journal Experts has offered English language help. This certificate was issued on October 31, 2019 and may be verified on the AJE website using the verification code ED97-710E-8BDE-F9F0-C8AB.
Junaid Ahmed (Reviewer 2): 1. A well written article but unfortunately it only highlights a technique and although its a new & novel technique, it would be desirable if it was conducted on patients and analysed for results in comparison to other techniques for a better understanding of its efficacy.

Response: Thank you for your appreciation. We introduced a modified method to dynamically present imaging results of the ocular fundus. Quantitative research compared to other techniques is ongoing, but it will take some time to complete, so we wanted to share this technology with you first. In this technical advance article, we mainly introduced the principle, method, and result of this technique.

2. Needs language correction at a few places.

Response: Thank you for your reminder. American Journal Experts has offered English language help. This certificate was issued on October 31, 2019 and may be verified on the AJE website using the verification code ED97-710E-8BDE-F9F0-C8AB.

Thanks for your consideration.

Best regards.

Sincerely Yours,

Qinghua Qiu